**Author’s response to reviews**

**Title:** Diagnosis and Antibiotic Treatment of Group A Streptococcal Pharyngitis in Children in a Primary Care Setting: Impact of Point-of-Care Polymerase Chain Reaction

**Authors:**

Arundhati rao (ari.rao@bswhealth.org)

Bradley Berg (Bradley.Berg@bswhealth.org)

Theresa Quezada (Theresa.Quezada@bswhealth.org)

Robert Fader (Robert.Fader@bswhealth.org)

Kimberly Walker (Kimberly.Walker@bswhealth.org)

Shaowu Tang (shaowu.tang@roche.com)

Ula Cowen (ula.cowen@roche.com)

Dana Duncan (dana.duncan@roche.com)

Joanna Sickler (Joanna.sickler@roche.com)

**Version:** 3  **Date:** 21 Dec 2018

**Author’s response to reviews:**

The authors thank the Editor for the rapid review of the revised manuscript.

Comment 1:

Response: We have added the manufacturers for the Quidel RADT, Roche NAAT and Quidel NAAT to the methods section. (page 6, lines 131-134)

Comment 2:

Response: State that Roche funded the study and is the manufacturer of the Liat in the declarations section (page 16, lines 326-327). Added that Roche is the manufacturer of Liat to the funding section (Page 17, lines 343-344)

Comment 4:
Response: Added Roche to the abstract (page 3, line 46)